Author's response to reviews

Title: Effect of obesity on constant workrate exercise in hyperinflated men with COPD

Authors:

Louis Laviolette (louis.laviolette@criucpq.ulaval.ca)
Francesco Sava (Francesco.sava@criucpq.ulaval.ca)
Denis E O'Donnell (odonnell@queensu.ca)
Katherine A Webb (kathy.webb@queensu.ca)
Alan L Hamilton (alan.hamilton@boehringer-ingelheim.com)
Steven Kesten (steven.kesten@boehringer-ingelheim.com)
François Maltais (francois.maltais@criucpq.ulaval.ca)

Version: 4 Date: 21 May 2010

Author's response to reviews:

Dear Editor,

Following your recommendations, we made the following changes to the manuscript.

1. Formatted the title, author names and email addresses
2. Removed shading from table
3. Specified figure titles in figure legends
4. Formatted abbreviations into paragraph

Thank you for your comments.

François Maltais